Author’s response to reviews

Title: Evaluation of Data Quality of interRAI Assessments in Home and Community Care

Authors:

Sophie Hogeveen (s2hogeve@uwaterloo.ca)
Jonathan Chen (x2chen@uwaterloo.ca)
John Hirdes (hirdes@uwaterloo.ca)

Version: 2 Date: 04 Oct 2017

Author’s response to reviews:

Dear Lingling Tian,

Thank you for indicating that my manuscript may potentially be acceptable. Since I first submitted, there have been some changes in the structure of the home care sector in Ontario. I would like to make a couple minor changes to the text to acknowledge these changes and make sure the manuscript is current. I have changed the following line on page 4, inserting the brackets seen here: "The Ontario Association of Community Care Access Centres (known as Health Shared Services Ontario as of 2017) receives and compiles data from each Community Care Access Centre (integrated into the Local Health Integration Networks as of 2017) in the province."

These are the only changes made. The document provided is a clean version.

Best regards,

Sophie Hogeveen